Author's response to reviews

Title: Tuberculosis suspicion and knowledge among private and public general practitioners: Questionnaire Based Study in Oman

Authors:

Abdullah A Al-Maniri (abdullah.almaniri@ki.se)
Omar A Al-Rawas (orawas@squ.edu.om)
Fatmah Al-Ajmi (alajmih@hotmail.com)
Ayesha De Costa (Ayesha.De.Costa@ki.se)
Bo Eriksson (bo@nhv.se)
Vinod K Diwan (vinod.diwan@ki.se)

Version: 4 Date: 15 May 2008

Author's response to reviews:

To the Editor: BMC Public Health

Subject: Re-submission of revision of manuscript entitled ‘Tuberculosis suspicion and knowledge among private and public general practitioners: Questionnaire Based Study in Oman’ for consideration in your journal.

We would like to thank you for accepting our paper for publication at your journal. In the enclosed new version of the manuscript we made all requested formatting changes suggested by the editorial production team. The changes made are mentioned below point by point.

Word and table counts - Please remove these.

Email address- please include email addresses for all the authors on the title page were included as suggested

Abstract – objective of the study is now in the background section of the abstract.

Keywords – Key words were removed.

Introduction – It is renamed to Background’.

Materials and Methods - It is renamed to ‘Methods’.

Authors contribution – Suggested changes were made in this section.

Ethical considerations – It is now included in the Methods section.

Funding – It is now included in the Acknowledgements section.

References - Table is removed.

Tables - Changes were made.
Additional files - All additional files are now mentioned below the tables.

Regards

*Corresponding Author:
Abdullah Al Maniri
Division of International Health (IHCAR), Department of Public Health Sciences, Karolinska Institutet
SE 17177 Stockholm
Sweden
Email: abdullah.almaniri@ki.se
Phone: +46737639144 Fax: +468311590